Author’s response to reviews

Title: Transperineal ultrasound-guided prostate biopsy is safe even when patients are on combination antiplatelet and/or anticoagulation therapy

Authors:

Kimitoshi Saito (k.saito@jichi.ac.jp)
Satoshi Washino (suwajiisan@jichi.ac.jp)
Yuhki Nakamura (yuhki.dreamskyward@gmail.com)
Tsuzumi Konishi (tsuzumi0203@gmail.com)
Masashi Ohshima (f_2_pk_gummy@i.softbank.jp)
Yoshiaki Arai (aria@nishiohmiya-hp.or.jp)
Tomoaki Miyagawa (sh2-miya@jichi.ac.jp)

Version: 1 Date: 28 Jun 2017

Author’s response to reviews:

Dear Editor

Thank you for your reply. We would like to submit the revised manuscript with a point-by-point response letter.

Editor

1. Please provide a statement in the Ethics approval and consent to participate section of the declarations. If the need for consent was waived by the ethics committee due to the retrospective nature of the study then please state this.

Response

According to the advice, we added a statement in the Ethics approval and consent to participate section of the declarations. (Declarations section, line 5, page 13).
2. Please include the tables in the main manuscript file rather than uploading them as separate files and place them at the end of the main text after the references.

Response

According to the comment, I included the tables at the end of the main text after the references.

3. When you have completed all the requested revisions please upload an entirely clean version of the manuscript with no revisions indicated.

Response

According to the advice, I uploaded an entirely clean version of the manuscript with no revisions indicated.

We thank you for your consideration and look forward to hearing from you.

Sincerely,

Kimitoshi Saito